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General Comments

CCO 314, SAMPLING AND TESTING A354 GRADE BD MATERIAL:

There is no work in the field on this operation today by ABF, but there is work by VGO. The Townsend
Test (Test IV) for TR’s 18 & 19 setup work is ongoing. ABF is working in the field today (but not at the Pier
7 warehouse area) on other operations this Saturday – ABF is working on the maintenance traveler. The
other operations are not covered by this diary.

VGO continues work on site today. From VGO, Dave Van Dyke and Pamela Wallace start work on site at
0800. After work yesterday to attach the TR 19 strain gauges to the wire runs and to install the TR 18
strain gauges, VGO works today to attach the TR 18 strain gauges to the wire runs. There are several QC
checks involved in the work today on the wiring. VGO completes high speed checks with the eDAQ to
look for issues with the instrumentation and will keep the eDAQ recording data over the rest of the
weekend for a low speed check to look for instrument drift. VGO leaves the site at approximately 1130
after completing all the field work they can do. VGO will monitor the data over the rest of the weekend for
issues and be back on site Monday morning.

Yesterday, VGO was complete with all planned turning of the rods for strain gauge installation and QC
checks. There was still the possibility that if the later QC checks revealed any problems, the rods would
need to be turned again. The AE sensors were installed by CT-METS on the couplers yesterday despite
the risk that they would need to be removed if the rods needed to be turned again. Today’s continued QC
checks by VGO do not reveal any issues that require turning the rods or removing the AE sensors.

ABF engineer Kelvin Chen is not at work today. ABF superintendent John Perine and ABF safety
manager Ed Fuqua are at work, but it is for other ABF work, not for the CCO 314 operations.

INSPECTOR OT REMARK:

Field and Office 6 hours: ABF is not working in the field on CCO 314. VGO’s shift is 0800 to 1130 in the
field. I am in the field part-time for VGO’s work and I am in the office the remainder of the time for various
work related to A354 Grade BD bolts and rods. My shift and OT are 0800 to 1430.
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